Author’s response to reviews

Title: A proposed method to investigate reliability throughout a questionnaire

Authors:

Tore Wentzel-Larsen (tore.wentzel-larsen@r-bup.no)
Tone M Norekvål (tone.norekval@helse-bergen.no)
Bjørg Ulvik (bjorg.ulvik@hib.no)
Ottar Nygård (ottar.nygard@helse-bergen.no)
Are H Pripp (are.hugo.pripp@oslo-universitetssykehus.no)

Version: 7 Date: 3 October 2011

Author’s response to reviews:

Dear Editor and production team;

We have made four minor changes to the text.
* At the end of page 6, the last part of a sentence has been changed from "...towards to end of the questionnaire..." to "...towards the end of the questionnaire...".
* In reference 8, 9 and 10 in the reference list, "jalowiec" is changed to "Jalowiec" since it refers to the name of the person.